## The Challenge of Incomplete Law And How Different Legal Systems Respond to It<sup>1</sup>

Katharina Pistor Columbia Law School

and

Chenggang Xu
Department of Economics, London School of Economics

Paper prepared for the Project Le Bijuridisme: Une approche économique

January 2004

 $^1\,Financial\,support\,by\,the\,Canadian\,Government's\,project\,on\,bi-juralism\,is\,gratefully\,acknowledged.$ 

1

#### 1. Introduction

Two hundred years ago France set out to codify its law in the Napoleonic codes. These codes covered what were then perceived to be the core areas of any legal system, whether it belonged to the common or civil law family: civil law, criminal law, commercial law, civil procedure and criminal procedure law. Courts were the primary institutions charged with enforcing these laws. Over the last one hundred years, however, in volume and arguably importance, these classic areas of the law have been outgrown by administrative or regulatory law, which encompasses everything from financial market over safety and food, to telecom regulation. In addition, there have been major changes in the institutions that administer and enforce this body of law. Law enforcement powers have been delegated to administrative bodies. Alternatively, new institutions, often referred to as "independent" regulators, were established to enforce this growing body of law. This development is often characterized as the rise of the 'regulatory state'.<sup>3</sup>

This paper interprets the rise of the regulatory state as a response to the deterrence failure resulting from incomplete law. In previous work we developed the incomplete law theory to explain the increasing importance of regulators in the most developed economies. In that work as well as in the current paper we focus on countries that use formal law as a major governance device, or that have made a firm commitment to the rule of law. Moreover, we limit the analysis to a particular sub-set of regulation: the management of risk when actions can be economically beneficial, yet carry the danger of

<sup>&</sup>lt;sup>2</sup> (Glendon, Gordon, and Osakwe, 1994). <sup>3</sup> (Glaeser and Shleifer, 2003; Landis, 1938; Moran, 2002).

<sup>&</sup>lt;sup>4</sup> (Pistor and Xu, 2003; Xu and Pistor, 2003).

causing substantial harm. Other areas of regulatory policies, such as regulation aimed at redistribution or other social goals, are excluded from our analysis.<sup>5</sup>

The incomplete law theory suggests that law is inherently incomplete and therefore cannot offer solutions for problems which are not precisely specified in the law. For the purpose of this paper we define law as a rule or standard that is established by a body with lawmaking authority rather than the legal system in a broader sense. In other words, we are referring to "loi" or "Gesetz" rather than "droit" or "Recht". The relevant lawmakers may choose to make law relatively more or less complete. Still, even if they were inclined to write a "complete" law, they would be bound to fail given law's inherent incompleteness at least when confronted with a changing environment. Incompleteness of law thus does not imply a negative connotation, but a simple fact.

An important implication of the insight that law is inherently incomplete is that law cannot effectively deter harmful actions at a level that would be socially optimal. We define the social optimum as a hypothetical world in which those actions (but only those) would effectively be deterred, the social harm of which exceeds its social and private benefits. When socioeconomic development and/or technological change is rapid, the incomplete law issue is more pronounced as legal development is not able to keep pace with these changes. This undermines the ability of any legal systems to effectively deter harmful actions, causing harm not only to individuals, but to public goods, including public health or the functioning of financial systems. The countries we analyze in this paper have all responded to the problem of deterrence failure caused by incomplete law by creating and allocating regulatory functions (for a definition see below) to state agents

<sup>&</sup>lt;sup>5</sup> For a detailed account of policy concerns that may justify regulatory intervention, see (Stewart, 1975) at pp. 1691.

other than courts. They have done so irrespective of whether they belonged to the civil or the common law family. These legal systems did not only amend existing and enact new law. Nor did they limit their reforms to changes in procedural law, for example by increasing the standard of care, by reallocating the burden of proof, or by extending standing in court. They created new agents and vested them with combined lawmaking and law enforcement powers. As a short hand, we refer to these agencies as regulators. Regulators have not replaced courts. In fact, as will be further discussed below, courts are frequently used as agents of governance over regulators. However, regulators have filled a gap a legal regime that relies entirely on reactive law enforcement by courts cannot fill. In this sense, courts and regulators are not substitutes, but complement each other.

The major difference between regulators and legislatures as lawmakers is that while the lawmaking powers of regulators is typically narrower than that of legislatures, regulators are less burdened by procedural requirements and can therefore more flexibly respond to changes. The major difference between regulators and courts as law enforcers is that courts have to wait for others to bring action before being able to enforce the law. By contrast, regulators are typically empowered to initiate enforcement procedures. Obviously, this characterization captures only basic functions of legal institutions and ignores variations across legal systems. However, we submit that these functions are crucial for understanding why regulatory functions have become necessary irrespective of a particular political constellation, which is often blamed for this phenomenon.<sup>7</sup>

-

<sup>&</sup>lt;sup>6</sup> As noted above, many specific forms of regulation are not covered by this study. Moreover, as will be further discussed below, regulators are not always separate institutions. In many countries regulatory functions are carried out by traditional state administrations, such as the ministry of health, agriculture, finance, etc.

<sup>&</sup>lt;sup>7</sup> This is most explicit in the US, where the rise of the regulatory state has been closely associated with the progressive administration of FDR Roosevelt and his "New Deal". For a historical account, see, for example (Sunstein, 1987).

All leading industrializing countries have allocated some lawmaking and law enforcement powers to regulators in some areas in response to the challenges posed by the socioeconomic and technological change brought about by the process of industrialization. While this has mitigated the enforcement problems posed by incomplete law, it has created problems of its own, and most importantly: how to control regulators. Governance structures had to be created that would not fall prey to the same incomplete law problems they were created to overcome. If law is inherently incomplete, as we suggest, then it is obviously impossible to control agency decision making by writing a complete law that would clearly set out their jurisdiction. Thus, other governance devices have to be developed to address the problem of governing regulators. The question "quis custodiet ipsos custodes?" is an ever-recurring governance problem. In this paper we point out that legal systems have toyed with different approaches to resolve the governance problems. Simply put, some use oversight and monitoring within a hierarchical system, others rely more on horizontal mechanisms, including inter-agency monitoring as well as litigation by affected citizens. Our major point here is that legal family alone provides very limited explanation power on evolution and governance of regulatory regimes. Instead, how lawmaking and law enforcement power is allocated may explain the evolution and governance of regulation better. Together with legal families, historical legacies, path dependency as well as constitutional constraints affect the choices for legal systems and may thus at times impinge their ability to optimize governance. By the same token, this also implies that governance structures that may work well in one particular legal system may create tensions when super-imposed on another. It should be the task of bijuridical analysis, and more broadly, of comparative

law, to analyze the "fitness" or alternative governance mechanisms in light of these constraints.

The paper is organized as follows. Section 2 summarizes our incomplete law theory. Section 3 discusses the initial allocation of lawmaking and law enforcement powers in traditional tort law. Section 4 describes how different countries have reallocated lawmaking and law enforcement powers to regulators. Section 5 presents a theoretical discussion of alternative governance mechanisms to control regulators. Section 6 explores particular regime choices made by different jurisdictions and their compatibility with existing constraints. Section 7 concludes.

#### 2. Incomplete Law and Deterrence Failure

The incomplete law theory we develop elsewhere<sup>8</sup> was inspired by the incomplete contract theory developed in the economics literature.<sup>9</sup> We define incomplete law negatively by contrasting it with an ideal type complete law. We regard a law as complete, if all relevant applications of the law are unambiguously stipulated in the law and the law can therefore be enforced literally provided that evidence is established. This requires that the law is self-explanatory, i.e. that every addressee agrees to the meaning of the law, and by implication that there is no need for a third party to interpret the law. Otherwise, a law is incomplete, that is, some of the relevant issues are not stipulated in the law or they remain ambiguous. An incomplete law cannot be enforced literally even when evidence is established.

<sup>&</sup>lt;sup>8</sup> See Pistor and Xu supra note 4.

Our basic premise is that law is intrinsically incomplete. <sup>10</sup> This is the case, because law is generally designed to last long and to apply to a large group of addressees and cases. These attributes contribute to the stability of the legal system and ensure equality before the law. At the same time, however, they invariably increase the incompleteness of law. Just as contractual parties cannot foresee all future contingencies, lawmakers (including legislatures and judges) have limited capacity to proscribe all possible issues that may arise under the law. To avoid that in each case where the law does not fit exactly a case will be dismissed or an accused acquitted, the power as to who may determine, if and how a law shall be applied to a particular case must be allocated, as otherwise the law cannot be enforced. We call this power to apply existing law to new cases and adapt it in the process of doing so the "residual" lawmaking power. The power to make new law from scratch, by contrast, is the "original" lawmaking power.

Note that failure to specify actions the law seeks to deter ex ante is different from establishing evidence ex post. In many cases, violators of the law will escape liability, because it is either impossible or too costly to establish evidence about who took the relevant action or the causal relation between the action and the harmful outcome – an issue that is well explored in the law enforcement literature. By contrast, incompleteness of law implies that it is impossible for the lawmaker to specify all relevant actions that may result in the harm the law seeks to prevent. This problem may be compounded by evidence problems, but is logically separate. When considering

9

<sup>&</sup>lt;sup>9</sup> (Hart, 1995). For a differentiation of the incomplete law theory from other theories, in particular the indeterminacy of the law, see (Pistor and Xu, 2003) supra note 4.

<sup>&</sup>lt;sup>10</sup> On the surface the notion of incomplete law may sound trivial or not completely new to the literature (e.g. H.L.A. Hart's notion of the indeterminacy of law shares some features of incomplete law (Hart, 1961)). The major contribution of this theory, as we see it, however, is to offer a rigorous framework for analyzing institutional design and the relation between legal design and institutional design.

whether an action should be litigated or prosecuted, lawyers typically consider the hypothetical case that evidence can be established to determine whether or not the relevant action can be challenged on legal grounds. If this is uncertain, the deterrence effect of the law is undermined. For most cases this is tolerable as the expected level of harm is relatively low. When the expected level of harm is sufficiently high, however, in particular when the number of victims is expected to be high and/or the action is likely to result in additional negative externalities, the inability of the law to effectively deter may cause substantial social welfare loss. Examples include the failure of the law to deter the use of harmful substances that may seriously affect health or life of many victims, or stock fraud schemes that may not only harm investors buying shares of a particular company, but undermine confidence in financial markets, which may ultimately result in market collapse with further consequences for the economy.<sup>12</sup>

If it was possible to specify ex ante the type of actions that may cause the harm the law seeks to prevent, law could be designed to deter harm caused by these type of actions and determine the appropriate level of liability or sanction to deter these actions. Thus, if the lawmaker could specify all harmful substances ex ante and determine in what cases the social harm outweighs any private or social benefits, courts could effectively enforce the law and most harmful actions would be deterred. Similarly, if lawmakers could anticipate all side effects of pharmaceuticals, they could write a law to ensure that only drugs were produced for which the side effects were manageable. 13 Finally, if it was possible to determine ex ante the many ways in which investors might be cheated by

 $<sup>^{11}</sup>$  (Becker, 1968; Stigler, 1970). For a survey of this literature, see (Polinsky and Shavell, 2000). (Romer, 1990).

<sup>&</sup>lt;sup>13</sup> Alternatively, the lawmaker may require mandatory insurance for the 'most dangerous' drugs, substances, etc.

share issuers, the law could be written to deter most of these actions – and so forth. Unfortunately, in many cases this is too costly, or simply impossible, because the possibilities brought about by scientific and technological change as well as the ingenuity of individuals in pursuit of maximizing their interests are impossible to foresee.

One may hold against this argument that it might still be possible to state that liability will be incurred whenever the social harm outweighs private plus social benefits. However, at best such a statement would be too vague to be of any useful guide to actors or courts, and at worst it may over-deter and thereby undermine social, scientific, and technological progress.

Traditionally, the power to determine the scope of a law when applying it to a particular case has been vested with the courts, both in civil and in common law jurisdictions. Increasingly, however, this power has been shifted to, or is shared with, regulators. A number of authors have already suggested that, under certain conditions, regulators may be better placed than courts to enforce the law at socially optimal levels. Some see the major reason for the superiority of regulators in their ability to *make* rules ex ante where courts are limited to ex post lawmaking. 14 Others have stressed that under conditions that resemble complex and long term contracts institutions that are capable of continuously monitoring or managing these relations are required. <sup>15</sup> We build on these contributions and suggest that a further important institutional feature of regulators is that they have the power to initiate enforcement procedures, whereas courts remain passive until another party (the victim or a state agent) has taken the initiative. We therefore call courts "reactive", and regulators "proactive" law enforcers. The fact that courts are

 <sup>14 (</sup>Shavell, 1984).
 15 (Goldberg, 1976; Priest, 1993).

reactive law enforcers enhances their independence and impartiality. By designing rules and initiating enforcement procedures, regulators take sides. Moreover, the combination of lawmaking and law enforcement powers in the hands of a single state agent gives regulators substantial discretionary powers. This raises new and important governance challenges.<sup>16</sup>

In our earlier work we show that when law is highly incomplete, reactive law enforcement by courts cannot ensure socially optimal levels of law enforcement. In particular, reactive law enforcement alone may result in deterrence failure and cause major harm to social goods, such as financial systems or public health. Allocating lawmaking and law enforcement powers to agents that can initiate enforcement proceedings and change rules flexibly to accommodate technological or socioeconomic development, can enhance the socially optimal level of law enforcement and thereby improve social welfare. Put differently, we argue that the invention of regulatory functions was at least in part determined by the inability of traditional law enforcement institutions to effectively cope with the challenges posed by socioeconomic and technological change. Other theories hold that regulators were created in response to interest group pressures and serve primary their needs. <sup>17</sup> Moreover, it has been suggested that legislatures often favor delegating lawmaking functions to regulatory agents in order

<sup>&</sup>lt;sup>16</sup> Governance of courts and judges is a difficult task as well. The debate about judicial independence and impartiality has long moved beyond the simple notion that independent judges are bound by the law and nothing but the law and will act in accordance with these constraints. On the debate of judicial independence (Landes and Posner, 1975; Posner, 1993); critically, however (Boudreaux and Pritchard, 1994). Further on the political economy of judicial independence (Ramseyer, 1994) and (Stephenson, 2003). While we recognize the importance of the question as to how judges are held accountable to the law, in the current research we focus on regulatory accountability not the least because of the extensive powers regulators have as a result of vesting combined lawmaking, adjudication and law enforcement functions in their hands.

<sup>&</sup>lt;sup>17</sup> See (Stigler, 1971) and (Posner, 1974) for these earlier theories of regulation.

to avoid political responsibility. <sup>18</sup> More recently, Majone has suggested that regulators are a device used by legislatures to achieve credible policy commitments. <sup>19</sup> Our theory differs from the above in that we suggest that the fundamental cause for regulation is not political, but is an inherent weakness of law when faced with conditions that render it incomplete and thus undermine its power to effectively deter harmful actions. We do not believe that politics are unimportant. In fact, political factors are, as we show in this paper, important determinants of regulatory regime choices. However, for the rise of regulators as proactive law enforcement agents, politics in our view plays role only to explain the differences among regulatory states, not the rise of regulatory states.

# 3. Initial Allocation of Lawmaking and Law Enforcement Powers in Common law and Civil Law Countries

Comparative scholars divide the law into two major legal systems – the civil law system and the common law system. Using our language of incomplete law and lawmaking power allocation, we can summarize the two systems as follows. A widely held notion in the literature is that the two systems differ fundamentally in the initial allocation of lawmaking powers. Civil law systems allocate original lawmaking powers to legislatures, and vest courts only with limited [CG: in the incomplete contract convention, if any party has all the residual control, that party will be the owner.

\_

<sup>20</sup> (David and Brierly, 1985; Zweigert and Kötz, 1998).

<sup>&</sup>lt;sup>18</sup> See (Fiorina, 1986). See also (Salzberger, 1993) for a similar argument for delegating lawmaking powers to courts.

<sup>&</sup>lt;sup>19</sup> This argument is in fact closely related to Landes and Posner's (supra note 16) argument about judicial independence. They argue that in order to ensure interest groups that bargains will be upheld even after they hand over power to the opposition, rulers will favor an independent judiciary.

Similarly here what we mean is that the court has some lawmaking power but not all] residual lawmaking powers (the scope of which varies across different areas of the law). In principle, courts may interpret the law, and may occasionally fill some gaps, but they may not make new law. In contrast, common law countries vest courts with important original as well as residual lawmaking powers.

This conventional wisdom is a gross over-simplification of how different legal systems function in the real world.<sup>21</sup> Not only is the boundary between interpreting the law, filling the gaps, and making new law often difficult to draw, but many provisions in, say the French or German civil codes, are so general that they implicitly allocate more extensive residual lawmaking powers to courts than the conventional wisdom suggests.

Rather than starting with the conventional wisdom, we determine the initial allocation of original lawmaking and law enforcement powers for specific areas of the law in the countries included in our study. They include England, France, and Germany, i.e. the "mother countries" of the major legal systems, the common law, the French and the German civil law systems. In our discussion of regulatory structures we also include the US and Canada. Both countries belong to the common law family, but as we will show below, their institutional designs differ not only significantly from each other, but — especially in the case of the US - in important ways also from England.

We start our analysis with simple tort law and ask how different legal systems have initially allocated the residual lawmaking and law enforcement powers in this area of the law. We select tort law, because this is the area of the law which traditionally has been responsible for allocating entitlements and risks between private parties that have no

<sup>&</sup>lt;sup>21</sup> (Mattei, 2000).

previous contractual relationship.<sup>22</sup> Much of the regulatory law we are interested in has complemented and sometimes superceded tort law.

Tort law is by nature highly incomplete. The type of actions that may result in harm, the level of damages that may be caused, and the assets that require protection by law are difficult to anticipate precisely by the original lawmaker. At the time the French Code Civil was enacted in 1804, railways, cars or airplanes had not been invented, nor could anybody have foreseen the massive changes technological progress brought to the production of goods, and the benefits, but also the harm that could result from these changes. In other words, the possibility of using DDT or Asbestos had not yet been explored, much less were the side effects from using these materials anticipated.

Given the incompleteness of tort law, major harm to society could result from underenforcement of the law, unless legal systems found ways to properly allocate residual lawmaking and law enforcement powers. Suppose a legal system enumerates the type of actions known that causes harm at the time the law is enacted and stipulates that only those actions can give rise to damages. Suppose further that this system denies courts the power of residual lawmaking and confines them to law enforcement. Such a system would most likely suffer from substantial enforcement failure problems. Not only would it be easy to circumvent the law by using actions not defined therein, but future harmful actions that were not anticipated at the time the law was enacted could not result in liability. Obviously, the lawmaker could change the law, but this takes time and in the interim much harm can be done. A legal system that used a sweeping definition of tort liability, but denied courts the power to exclude actions that do not result in major harm from liability would be equally problematic. Such a system would over-deter as courts

<sup>&</sup>lt;sup>22</sup> (Coase, 1960).

would have to impose liability even when only little harm was done, or when the causal relation between actions and consequences was only remote.

As will become apparent in our analysis, each legal system has developed a different mix between allocating extensive lawmaking powers to the courts and vesting these powers primarily with the legislature. The results are sometimes counterintuitive, at least if one starts from the simplified premise that civil law is rigid and constraints judges in lawmaking, whereas common law vests them with extensive lawmaking powers. As it turns out, France has the most sweeping definition of what amounts to a tort action that may result in harm and leaves it to the judges to delineate actions that should not be covered by it. As a result, tort law in France is primarily case law. By contrast, England has only gradually rid itself of a rather rigid framework of different tort actions, which courts had developed over time. Moreover, courts have increasingly deferred decisions about extending tort liability to the legislature and exercised remarkable judicial constraint in this respect.

#### United Kingdom

In the UK, the classic common law country, courts exercise both original and residual lawmaking powers to this day, even though statutory law is much more prominent today than it was two hundred years ago. Tort law can be described as the sum of all classes of cases that have been recognized by the courts as causes for liability absent a contractual relation.<sup>23</sup> Historically, a special writ had to be filed for each case. Special writs, or procedural forms to present a case, existed for conversion, nuisance, defamation, negligence, or deceit. Prior to the judicial reforms of 1875, failure to properly classify a

case could result in dismissal of the case. Even though this formalist approach was then abandoned, common law courts have long continued to classify tort cases in accordance with the traditional typology.

This rather narrow procedural approach created by the English common law may result in under-enforcement and thus deterrence failure, as existing categories did not always cover new cases that arose with the process of industrialization and socioeconomic development.<sup>24</sup> However, courts could, and indeed did adapt existing tort principles to new cases. Courts have developed liability for negligent conduct as a separate cause of action during the course of the 19<sup>th</sup> century. In response to the growing number of cases of accidents that resulted from the use of machines or the railways, courts held that not only positive wrongful actions, but also failure to follow reasonable standards of precaution could give rise to liability. In doing so they could uphold the traditional doctrine of "no liability without fault" while taking account of a changing environment in which individual action could place many others at risk.<sup>25</sup>

With the growing importance of the parliament as statutory lawmaker in the UK, courts have deferred what they considered to be substantive changes in the law to the legislature. In Derry v. Peek, a landmark case concerning the scope of liability of company directors for misrepresentation of information in a prospectus for the issuance of shares, the House of Lords explicitly noted that it may be desirable to hold directors to

22

<sup>&</sup>lt;sup>23</sup> See (Zweigert and Kötz, 1984) pp. 342 and pp 417.

<sup>&</sup>lt;sup>24</sup> For the inability of courts to respond effectively to the growing number of white collar crimes in late nineteenth century Britain, see (Robb, 1992). On the incompleteness of the case law that was created in response to these challenges, see Pistor and Xu (2003) supra note 4.

<sup>&</sup>lt;sup>25</sup> Obviously, courts did not develop their response to these new challenges in isolation. They could rely on legal doctrine (as pointed out by Antoniolli and Mattei in their comments to this chapter) as well as on a growing public debate on the effects of industrialization on morals, the social fabric and the legal system. See only (Banner, 1998) for an analysis of the cultural and political background against which financial market regulation developed in the United Kingdom and the United States.

a higher standards of care than required by classic fraud law, but that it would be for the legislature to take such actions. <sup>26</sup> In fact, the legislature intervened and enacted the Directors' Liability Act only a year after this decision came down, which in effect shifted the burden of proof that they acted properly to the directors.<sup>27</sup>

Where the legislature did not intervene, courts only cautiously extended the scope of liability. Thus, courts upheld the principle that negligence would result in liability only, if the injuring person owed some duty to the injured person. In a decision rendered in 1893, Lord Esher held that "A man is entitled to be as negligent as he pleases towards the whole world if he owes no duty to them". 28 Still, for goods that were deemed particularly dangerous, including poison, explosives, and weapons, a special duty was not required. Furthermore, in 1932, the House of Lords decided that the producer of goods could be held liable vis-à-vis the final user, to whom he, in principle, did not owe a special duty, if it was impossible for the retailer or the final user to ensure that the good did not suffer any defects.<sup>29</sup> Nevertheless, strict liability, i.e. liability without negligence or intent, has been acknowledged by English common law courts only in exceptional cases. 30 In principle, courts have called for legislative intervention and an increasing number of special laws, including the Civil Aviation Act of 1949, the Nuclear Installations Act of 1965 and the Gas Act of 1965 establish strict liability for harm caused in the operation of these facilities. These acts, however, are highly incomplete in that each covers only a

<sup>&</sup>lt;sup>26</sup> Derry v. Peak (1889, 14 A.C. 377) at p. []. See also our analysis of this case in (Pistor and Xu, 2003).

<sup>&</sup>lt;sup>28</sup> Lord Esher in LeLievre v. Gould, [1893] , 1 Q.B. 491, 497 as quoted in (Zweigert and Kötz, 1984) at

 <sup>&</sup>lt;sup>29</sup> See Donoghue v. Stevenson, [1932] A.C. 562.
 <sup>30</sup> Arguably, only a single case has been handed down that establishes strict liability, namely Ryands v. Fletcher [1868], LR. 3 H.L. 330, which involves a case where a an owner who built a water reservoir on his property was held liable for damages caused when the water flooded the coals mines on the adjacent

particular industry. Industries or services not covered by special legislation are still covered by the traditional common law of torts.

Interestingly, courts in the United States, albeit also a common law country, have been more aggressive in establishing strict liability at least in cases where a product or the operation of a facility is deemed inherently dangerous. In the US the legislature is increasingly called upon to intervene in order to *limit* rather than expand the liability of producers or operators.<sup>31</sup> This example already suggests, that legal families may have only limited explanatory power for explaining the scope of lawmaking powers courts are willing to assume or the deference they grant to either legislatures or regulators.

#### France

In France, the original lawmaking powers of the courts were curtailed with the codification of core areas of the law at the beginning of the 19<sup>th</sup> century and, as will be further discussed below, with explicit constraints placed on the judiciary's ability to review executive or administrative acts. Nevertheless, courts have continued to play an important role in residual lawmaking. In fact, the civil code implicitly allocates substantial residual lawmaking powers to the courts by including a very broad general tort provision, which establishes that anybody who causes harm to another either intentionally or negligently, is obliged to compensate.<sup>32</sup> Given the broad mandate in favor of liability established by the code, it is not surprising that courts had less trouble than

property, even though negligence could not be established. The major reason for granting liability was that the property had been used in a non-natural fashion [CHECK].

<sup>&</sup>lt;sup>31</sup> The most recent example is a bill aimed at limiting the liability of gun producers [CHECK].

<sup>&</sup>lt;sup>32</sup> The two leading provisions are Arts. 1382 and 1383 of the code. Art. 1382 reads, "Tout fait quelconque de l'homme, qui cause à autrui un dommage, oblige celui par la faute duquel il est arrivé, à le réparer. Art. 1383 reads, "Chacun est responsable du dommage quail a causé non seulement par son fair, mais encore par sa négligence ou par son imprudence".

either English or German courts to recognize liability for immaterial damages (which German courts deny), or, in the case of wrongful deaths, to hold the injurer liable to large group of claimants, including not only the immediate dependants, but also siblings, foster parents, or fiancées, and in individual cases even the employer of the deceased.<sup>33</sup> The highly incomplete nature of the law has resulted in cases being referred to the French supreme court (cour de cassation) to address the question, whether even the creditor of the deceased should be allowed to claim compensation from someone who caused the debtor's death and was unable to obtain relief from his legal successors. Thus, it has been the task of the French courts [CG: did you mean the supreme court? If not, I am a bit confused. If yes, then there is a typo?] to constrain the scope of the sweeping liability established in the code, rather than to expand it gradually, as has been the case in England.

An important condition for liability under French law is fault in the form of either negligence or intent. Again, courts have not had much trouble in recognizing fault in a diverse set of cases, including for misuse of rights (*abus d'un droit*), which has given rise to liability in cases where contracts were dissolved in a manner that caused damages to the other party. As far as strict liability is concerned, the code itself explicitly requires negligence or intent and until the end of the 19<sup>th</sup> century it was virtually undisputed that liability required fault. In 1896, however, the *court the cassation* handed down a decision in which it presented a new interpretation of the tort provisions contained in the civil code. Several provisions of the code establish liability for harm caused by an animal or building in the possession of or under the guard of a person, who *qua guardianship* is

<sup>&</sup>lt;sup>33</sup> (Zweigert and Kötz, 1984) at p. 359. For example, a soccer club could demand damages from the person who caused the death of a soccer player for the transfer sum they had to pay.

held liable for harm caused by these objects or things. The court extended the application of these provisions to other cases where things under guard ("que l'on a sous sa garde") caused the damage and ruled that such circumstances gave rise to the presumption that the person was liable. A growing body of case law to this end was confirmed in 1930 by the famous "arret Jand'heur", which established the general presumption that someone deemed guardian of a thing (i.e. in this case a car), could be held liable for any harm caused by it regardless of fault.<sup>34</sup> The court thereby established strict liability of the owner of a car for damages caused by it. France has also enacted a number of special laws that create strict liability for particular industries. Nevertheless, case law is the major source of tort law in France in the sense that it is the most important device for delimitating those actions that should not incur liability notwithstanding the broad definition of tort liability in the Code Civil.

#### Germany

The German civil code was enacted only in 1900 [CG: did you mean 1800?] and presents the culmination rather than the starting point of legal responses to the challenges of the 19<sup>th</sup> century. Some trends that proved important for the further development of tort related law originated prior to the adoption of the civil code. In particular, Prussia established strict liability for railway accidents by passing a special statute on this matter. Moreover, in 1871 the law on liability (*Haftpflichtgesetz*) was enacted, which establishes

<sup>&</sup>lt;sup>34</sup> Ch. Réun. 13.2.19930, S. 1930. 1. 121.

strict liability for all damages to life or health that occur in relation to the operation of a train (*bei dem Betriebe*).<sup>35</sup>

The German civil code (BGB) includes a handful of tort provisions. The general tort provision is less sweeping than its French counter part.<sup>36</sup> Only damages caused to life, liberty, health or property can give rise to liability. There is a small window for expanding the scope of the law, as the code explicitly mentions "other rights" (*sonstige Rechte*). Much of the early case law was therefore devoted to establishing *what* other rights might be recognized as similar to those enumerated to give rise to liability.<sup>37</sup>

Just as in France, the code itself establishes liability for intent as well as negligence. Yet, German courts have been more reluctant than French codes to develop the scope of liability further and impose strict liability by way of case law. Arguably, this avenue was already precluded by specific legislative interventions, such as the 1838 Prussian railway law, or the 1871 general liability law. In this case, the legislature had indicated its original lawmaking powers in this area and courts respected this. As a result, strict liability to this day is confined to areas where the legislature has enacted a special statute establishing strict liability. In most cases these statutes also include upper ceilings for the amount paid for a single accident.<sup>38</sup>

<sup>&</sup>lt;sup>35</sup> A revised version of the same law is still in force today. See *Das Haftplichtgesetz* as (re-)announced on 4 January 1978, BGBl.I, 145) and last amended on It now includes liability for damages to life or health of persons that result from the use/operation of electricity, gas, smoke or liquid, as long as the damages are caused by a facility that is connected to their use.

caused by a facility that is connected to their use.

36 Section 823 of the Civil Code (last re-enacted 2 January 2002, BGBl I, p. 52 with further amendments introduced on 15 December 2003, BGBLI, 2676) states that whoever negligently or intentionally damages someone's life, liberty, or property will be liable for compensation. "Wer vorsätzlich oder fahrlässig das Leben, den Körper, die Gesundheit, die Freiheit, das Eigentum oder ein sonstiges Recht eines anderen widerrechtlich verletzt, ist dem anderen zum Ersatz des daraus entstehenden Schadens verpflichtet."

37 For details, compare (Zweigert and Kötz, 1984) at pp. 340.

<sup>&</sup>lt;sup>38</sup> Ibid at p. 404.

#### Summary and Comparison

To summarize, throughout the 19<sup>th</sup> century, courts were the primary agents to determine liability for tort actions. The nature of the cases that were litigated changed considerably during the course of this century. Courts in the UK, France and Germany recognized the need to adapt existing law to the new cases that arose as a result of technological change. This was done on the basis of the original lawmaking powers in the case of the UK and the residual lawmaking powers that were implied in statutory law, in the case of Germany and France. The scope of implied lawmaking powers differs somewhat from country to country. As explained above, France had the most general provision on tort liability. By contrast, in Germany tort liability is limited to damages of enumerated rights, although the code adds that "other rights" (sonstige Rechte) may also be protected. This opened the backdoor for protecting a range of rights, which the original lawmakers could not anticipate – even though the fact that such a broad term, which invited judicial interpretation was included in the law suggests that lawmakers were only too aware of their own limitations. In England the classification of torts into highly specified categories gives the appearance that the scope of tort law as well as the ability of courts to respond to a changing environment was rather limited. In fact, courts could and did change the types of cases they would hear, and thereby opened up tort law to new cases as they arose.

The major difference between the three countries is that France started off with the general presumption in favor of liability, at least where fault could be established. And with the recognition of guardianship as a cause for liability, the way towards strict liability had been paved. By contrast, English courts started with the presumption of no

liability, unless the injurer had breached some duty he or she owed to the victim. This principle was only gradually relaxed over time. Finally, German courts had substantial residual lawmaking powers, but less so than their French counter part. In light of the extensive legislative interventions that had preceded the enactment of the civil code, however, courts in Germany have been more deferential to the legislature than courts in France. Table 1 below summarizes these major differences.

Table 1: Allocation and Use of Lawmaking Powers in General Tort Law

	lawmaking re	Allocation of residual lawmaking powers	Formal constraints on courts' lawmaking powers	Agent introducing major legal change	
	powers			Negligence	Strict liability
England	Courts Legislature	Courts	Precedents and legislative prerogative limit court's lawmaking powers	Courts	(Courts) Legislature
France	Legislature	Courts	Broad code provisions establish few formal constraints	Legislature	Courts Legislature
Germany	Legislature	Courts	Fairly broad code provisions establish some constraints	Legislature	Legislature

Source: Compilation by authors

These differences notwithstanding, the above analysis suggests that while courts in all countries proved to be quite responsive to socioeconomic change, ultimately they were unable to effectively protect society from harm. All countries suffered a series of financial market failures and watched the number of those injured at work, on streets, or rails increase. While it might be possible to identify particular weaknesses in the court system of one of the three countries that made them less apt to respond to this challenge, <sup>39</sup> we propose that the real problem runs deeper. All three countries have designed courts as *reactive* law enforcers. Their power to enforce the law, as well as to adapt it to change depends on others bringing action. By implication, most cases are brought only ex post, i.e. after harm has been done. Moreover, parties may hesitate to bring action when the outcome is highly uncertain, as is invariably the case when law is highly incomplete – i.e. when cases arise that have not been adjudicated before. As a

<sup>&</sup>lt;sup>39</sup> See, for example (Robb, 1992) for a critique on how English courts handled white collar crimes; and (Glaeser and Shleifer, 2003) who suggest that the rise of the regulatory state in the US occurred in response to high levels of corruption in the court system.

result, existing tort law could not effectively deter harmful actions. The major response to this problem was the emergence of agents that combined proactive law enforcement powers with residual lawmaking powers.

### 4. Allocating Lawmaking and Law Enforcement Powers to Regulators

The extent to which modern states have allocated lawmaking and law enforcement powers to agents with regulatory functions is remarkable. Describing the situation in the UK as late as 1950, Michael Moran notes that "regulatory standards covering the production and sale of food were skeletal" and while much economic life was regulated, it was regulated by self-regulators, such as the free profession, or the City of London. By contrast, today "vast new areas of social and economic life have been colonized by law and by regulatory agencies. The food we eat, the physical conditions we work under, the machines and equipment we use in our home, office and on the road – all are increasingly subject to legal controls, usually administered by a specialized agency". 41

A similar story can be told for Germany and France as well as for other highly industrialized nations. The scope of safety and system regulation greatly expanded in all industrialized countries especially in the period following World War II. A plausible explanation for this trend is the risk created by the process of industrialization and development on the one hand, <sup>42</sup> and the difficulties to deter all potentially harmful actions on the other (Pistor and Xu, 2003; Xu and Pistor, 2003). Frequently, the creation and

<sup>&</sup>lt;sup>40</sup> [Author?](2001) To be sure, our definition of regulators includes self-regulators and the emergence of a large number of self-regulator organizations in areas that elsewhere were regulated by the state, may suggest a response to a similar problem, even though it differs in institutional detail.

<sup>41</sup> ibid at p. 20.

allocation of proactive law enforcement powers followed a series of scandals, devastating accidents, or economic shocks. This is well documented for financial market regulation, which is typically introduced on the heels of scandals that cause major market downturns or even the (temporary) collapse of the market. The best example is the federal securities regulation in the US, which followed on the heels of the 1929 stock market crash. 43 More recently, the Sarbanes Oxley Act was a response to the law enforcement problems exposed by corporate scandals like Enron, Worldcom, Adelphi, etc..<sup>44</sup> Similar events occurred in other countries. In England, the parliament threatened on several occasions to regulate financial markets in the 19<sup>th</sup> century and again in the 1920s, unless stock exchanges as the major sell-regulators increased their regulatory oversight. 45 While this threat did not materialize until 1986 with the "Big Bang" reforms, it strengthened the regulatory oversight of companies by self-regulatory bodies. And in Germany, the first comprehensive state regulation of stock markets, the 1896 stock exchange law, was a direct response to the crash of the founders' boom. 46 It allocated regulatory oversight to relevant state agents at the level of the Länder (German states) and charged stock exchanges with scrutinizing companies before listing.

Similarly, the expansion of regulatory controls over hazardous chemicals in all major industrialized countries can be attributed to a series of disasters that occurred in the late 1960s and early 1970s, such as Soveto (Italy), Hopewell (Virginia, US), and Minamata (Japan).<sup>47</sup> Regulatory responses typically included approval requirements for the use of

40

<sup>&</sup>lt;sup>42</sup> (Beck, 1992; Giddens, 1990)

<sup>43 (</sup>Seligman, 1983); (Coffee, 1984).

<sup>44 (</sup>Bratton, 2002).

<sup>45 (</sup>Michie, 1999).

<sup>46 (</sup>Merkt, 1997).

<sup>&</sup>lt;sup>47</sup> (Brickman, Jasanoff, and Ilgen, 1985).

new and potentially dangerous substances, the regulator's power to enjoin the production or use of particular chemicals and to amend the list of hazardous substances over time.

As the above examples suggest, a frequent response to crises that exposed the inability of existing law to effectively deter actions that could result in major harm to the public was the allocation of lawmaking and law enforcement powers to agents, which unlike courts could enforce law proactively. Proactive law enforcement entails the power to initiate law enforcement procedures independently of others. This includes not only the power to enjoin a specific action before harm has been done, but to impose entry barriers in the form of registration and approval conditions on a particular class of actions that are deemed potentially dangerous. To give proactive law enforcement sufficient bite, they are typically combined with the power to adapt existing rules to changes in the environment. Thus, regulators can adjust disclosure or merit rules companies must meet to enter the market, or change the list of potentially hazardous substances.

In Europe the growth of regulatory functions did not imply the creation of new regulatory agents to the same extent it did in the United States. Unlike in the United States, where the rise of the regulatory state is – rightly or wrongly<sup>48</sup> - closely associated with the creation of independent regulatory agents, in Europe until very recently<sup>49</sup> regulation was placed in the hands of the executive branch of government. This is evident, for example, in the area of environmental and safety regulation of hazardous chemicals. Table 2 lists the major agencies in charge of controlling chemicals in the UK,

-

<sup>&</sup>lt;sup>48</sup> For a critique of this over-simplistic view, see (Strauss, 1984). In his words, "regulatory and policymaking responsibilities are scattered among dependent and executive-branch agencies in ways that belie explanation in terms of the work agencies do", ibid at p. 584.

<sup>&</sup>lt;sup>49</sup> The rise of the regulatory state has become subject of extensive scholarly analysis and debate only since about the mid 1990s. See, for example, (Majone, 1994).

France, Germany, and the US the scope of their regulatory powers and the date when these regulatory powers were created or expanded.

Table 2: Regulatory Functions and Agencies for Controlling Chemicals

	Regulatory Agency	Scope of Regulatory Oversight	Date when established/ regulatory powers conferred	
France	Ministry of Agriculture	Food additives Pesticides	1977 Chemical Notification Law	
	Ministry of the Environment	Notification of the use of toxic substances		
Germany	Ministry of Agriculture	Pesticides; efficacy testing and registration	1971 Environment Program launches series of legislative acts expanding regulatory oversight	
	Health Ministry	Regulation of pesticides, residues, new food additives		
	Federal Health Office	Toxicological evaluations; advise to federal health and agricultural ministry	1980 Chemicals Act	
England	Ministry of Agriculture, Fisheries and Food	Food additives, pesticides	Established end of 19 <sup>th</sup> century Regulatory responsibilities expanded in 1970s	
	Health and Safety Commission	Pre-market testing of new chemicals	Established 1974	
USA	Ministry of Agriculture	Regulation of pharmaceuticals	1906 Pure Food Drug Act	
	Food and Drug Administration		1938 Federal Food, Drug and Cosmetic Act	

Source: Compilation by authors on the basis of (Brickman, Jasanoff, and Ilgen, 1985) and (Temin, 1992).

This general pattern is currently undergoing important change. A growing literature on the rise of the regulatory state in Europe suggests that many countries are now moving to creating new, independent regulators. <sup>50</sup> In part this trend may be motivated by the harmonization of law at the European level and the greater emphasis placed on the coordination among national regulators. In order to facilitate this process, increasing attention is being paid to streamlining regulatory structures, regulators' competences, and enforcement powers. This has gone hand in hand with making more detailed institutional

<sup>&</sup>lt;sup>50</sup> (Majone, 1994; Thatcher, 2002).

demands on the design of such regulators in European harmonization directives.<sup>51</sup>
However, the jurisdiction of the EU over the design of national institutions is rather limited. In fact, sufficient differences remain across member states in the creation of "independent" regulators to suggest that EU sponsored reforms is only part of the story.<sup>52</sup>
Whatever the causes for the recent trend, the creation of "independent" regulators will require different governance structures from those that so far have dominated the European landscape. In the following section, we present a stylized account of alternative governance structures for regulators, which will then be used to analyze real world governance structures in different jurisdictions.

#### 5. Governing Regulators

The notion that law is incomplete and therefore unable to effectively deter harmful actions at socially optimal levels helps explain why there is a need for regulators and what its major functions are. We now address the question of the governance structure over agents that combine proactive law enforcement with delegated lawmaking powers, i.e. the issue of regulatory regime choice. We distinguish two ideal types of regulatory governance structure: a vertical and a horizontal one. In the vertical governance structure agents carrying out regulatory functions are accountable to their superiors, who in turn are accountable to whoever appointed or elected them. By contrast, the horizontal governance structure lacks these clear lines of accountability. Instead, governance is effectuated by a variety of mechanisms, including restrictions on the scope

<sup>&</sup>lt;sup>51</sup> A recent example is the draft Market Abuse Directive, which requires that national regulators have the power to impose administrative sanctions. COM(2001) 281 final. 30 May 2001.

of regulatory functions or approval requirements, public participation and transparency, and extensive judicial review.

The two governance structures are akin to the debate in economics about the determinants of vertical integration and market transactions.<sup>53</sup> According to transaction cost theory, vertical integration occurs when the internalization reduces transaction costs, in particular when parties to the transaction make highly transaction cost specific investments.<sup>54</sup> Further closer to our analytical framework, hierarchy or vertical integration vs. markets or non-integration are analyzed as results of allocating property rights when contracts are incomplete. When parties are incapable of stipulating all future contingencies in their contract, it is necessary to determine who holds the right to decide future conflicts, i.e. who holds the residual rights of control. Should they be independent to each other (the case of markets) or should they be put into a hierachical relationship (the case of integration)<sup>55</sup> The parallel to the governance structure of regulators is that when law is incomplete it is not possible to design effective governance structures that rely primarily on specifying regulatory tasks. Instead, additional governance mechanisms are needed. They may comprise of monitoring within hierarchy – the parallel to the integration of firms - inter-agency monitoring or competition, vesting individuals affected by regulators actions with the right to judicial recourse, and the like. A word of caution when applying theories of the firm in this context is however, in order. There are important differences between economic transactions and the design of legal institutions. In economics, incomplete contract theory (with some additional conditions) goes a long

<sup>&</sup>lt;sup>52</sup> (Gilardi, 2002).

<sup>53</sup> Williamson, Markets and Hierarchies, 1975.

<sup>&</sup>lt;sup>54</sup> (Williamson, 1979),

<sup>&</sup>lt;sup>55</sup> (Hart. 1995).

way in explaining when integration is optimal and when it may be better to use armslength contracting. By contrast, to explain the governance of regulators in different countries we need to employ not only incomplete law theory, but understand the institutional, including constitutional constraints of each system. The reason is that in economics the two contracting parties are assumed to have free choice over designing their contract and allocating powers among them – or to decide that vertical integration might be superior to arms length transacting. Third party repercussions are non-existent or are assumed away. By contrast, the design of law as well as of lawmaking and law enforcement institutions not only has important third party effects, but can influence and/or be influenced by the process of political decision making. It is the function of the constitution to limit choices over the content of law and, perhaps even more importantly, over the allocation of the power to make, adjudicate, and enforce the law. These constitutional constraints are binding and cannot be freely re-negotiated, even though the precise boundaries of the jurisdiction of different branches of the government may be reinterpreted over time. <sup>56</sup> Still, the parallels between firm level governance and regulatory governance helps identify major features of alternative regulator regimes.

The ideal type of a hierarchical governance resembles an integrated firm. New functions, in this case proactive law enforcement and complementary lawmaking activities are created inside government, i.e. inside existing ministries or administrations. They might be fully integrated, or anew department with specified regulatory functions might be created. The bureaucrats in charge of regulation are subjected to similar lines of control as other departments, in particular to oversight by the command center, the

<sup>&</sup>lt;sup>56</sup> On an interpretation of the US constitution that is consistent with the basic principles established in that document, yet takes account of the rise of the regulatory state, which had not been foreseen by the founding

government. The major line of defense by those subjected to regulation is some kind of review either by a special body within the relevant state bureaucracy, or by an independent court. While the former may have the advantage of greater expertise in the subject area, the latter offers more degrees of independence.

An alternative regulatory structure is to create new regulatory agents outside existing state administrations and to employ primarily horizontal mechanisms for governance purposes. They will explicitly not be subject to hierarchical lines of command and control. While this guarantees their independence and might make the combination of lawmaking and law enforcement functions within a single state body less problematic, it creates new governance problems. One governance device a lawmaker contemplating the delegation of lawmaking and law enforcement powers to an independent regulator might employ is to write a law that sets forth specifically the regulatory tasks. In writing this law, however, the legislature faces exactly the same incomplete law problem it confronts when writing any other law. Legislatures do not have unbounded foresight and will therefore inadvertently write an incomplete law. Incomplete law can result from too much specificity as well as from too much ambiguity. Specificity eases the monitoring and enforcement tasks, but reduces the ability of the regulator to adapt to a changing environment without seeking approval from the legislature. It also invites strategies to circumvent regulations. Ambiguity enhances the regulator's adaptability, but weakens monitoring and oversight by the legislatures, as well as other agents the legislature may employ, such as judicial review. Moreover, ambiguity raises the question or who holds the ultimate residual lawmaking powers to determine the proper scope of regulatory

fathers, see Strauss (1984), supra note [].

activities – the regulators or the courts.<sup>57</sup> Thus, the law that creates a regulator or "specifies" regulatory tasks is hardly sufficient as a governance device. Additional, external control mechanisms need to created to reign in regulators should they expand their powers too much.

Each regime has its own costs and benefits. While the hierarchical governance structure may give the executive better control over the administration of regulatory tasks, it thereby vests the executive with substantial powers. Moreover, hierarchical controls often lack transparency and their effectiveness very much depends on the quality of the government in power. Furthermore, with the growing number and complexity of regulatory tasks that are amassed by a single bureaucracy, monitoring and oversight within hierarchical systems becomes difficult and costly and relevant state agents might lack the incentives to perform adequately. A typical response observed in the private sector would be a spin off of some operation. Arguably, the recent trend in the European systems to create a number of so-called "independent" agencies can be interpreted to resemble such spin-offs.

By contrast, in lieu of effective internal controls the ideal type horizontal system has to rely to a greater extent on external governance mechanisms. One would therefore expect a greater role of the judiciary in ensuring accountability of regulators. In fact, it has been suggested in systems with independent regulators there is more extensive judicial control, but also greater participatory rights and oversight by citizens.<sup>59</sup> Citizen

-

<sup>&</sup>lt;sup>57</sup> This issue has been subject of substantial debate in the US after the Chevron decision of the Supreme Court. For a more detailed discussion, see below [X].

<sup>&</sup>lt;sup>58</sup> On a critique of the German internal governance model precisely for its lack of transparency and citizens' participatory rights, see (Rose-Ackerman, 1995). For a critical review of this position compare [Peter Lindseth ADD].

<sup>&</sup>lt;sup>59</sup> (McCubbins, Noll, and Weingast, 1987).

participation may also help address the problem of regulatory under-performance. There is obviously room for intrinsic motivation of bureaucrats in either system.<sup>60</sup> Yet, the right balancing between overzealous bureaucrats on the one hand, and too much slack on the other, might require some externally imposed constraints.

In the following section we will use this framework to analyze real world regulatory regimes. As will become apparent, none of the systems fits perfectly into one of the two ideal types. Interestingly, the resemblance with these ideal types was greater in the past, but has been weakened, as both systems have responded to the particular problems their initial regime choice posed. For the hierarchical system this has been the weakening efficacy of the internal governance structure given the growing number and complexity of regulatory tasks. For the horizontal system it has been the inherent weakness of legal controls caused by incomplete law.

#### 6. Comparing Regulatory Regimes

We begin our analysis with a summary of the regulatory regimes in the three jurisdictions discussed above: the UK, France, and Germany. All three countries have traditionally used regulatory regimes that are fairly close to the hierarchical model. Canada follows the British system more closely than does the US, which is more of an outsider among the countries surveyed. In the other four countries regulators are controlled indirectly by the legislatures through its control over government. Courts

<sup>&</sup>lt;sup>60</sup> For an extensive discussion of intrinsic and extrinsic motivation and a review of the empirical literature on the effect of , compare (Oserloh and Frey, 2000). See also (Landis, 1938) who points out that the danger of combining lawmaking and law enforcement functions in the hands of a single regulator may be countered by hiring a cadre of expert professionals and fostering a culture of professionalism.

exercise judicial review, but have limited their review mostly to questions of procedure and jurisdictional question and deferred substantive issues largely to the expertise of regulators. By contrast, the US system is the primary example for a horizontal regulatory regime structure. Increasingly, however, it has complemented external controls over regulators, including legislative oversight and judicial review with hierarchical controls exercised by the president and oversight bodies appointed by him. Table 2 summarizes the major differences in the governance of regulators in these five countries.

**Table 2: Regulatory Governance in Comparative Perspective** 

Tuble 21 Hegulatory Governance in Comparative Leispective						
Countries	Regulatory	Regulatory	Enforcement			
	Regime Choice	Discretion	Mechanism			
UK	Ministerial	Broad delegated	Judicial review by			
	bureaucracy &	powers	general courts limited to			
	Self-regulation		ultra vires review			
France	Ministerial	Broad rulemaking	Extensive review by			
	bureaucracy	powers granted by	Conseil d'Etat			
		constitution				
Germany	Ministerial	Relatively narrow	Review of violations of			
	bureaucracy	delegation of	individual rights by			
		lawmaking powers	administrative courts			
US	Independent	Initially restricted	Extensive judicial			
	regulators &	Expanded over time	review by general courts			
	executive agencies					
Canada	Ministerial	Broad delegated	Limited judicial review			
	bureaucracy	powers	by general courts			

#### United Kingdom

The UK has a tradition of allocating regulatory tasks to part of the existing state bureaucracy, which in turn is overseen by the government accountable to parliament. The creation of independent regulatory agencies is only a development of the last two decades. A description of the British regulatory system would, however, be incomplete without taking into account the scope of self-regulation. Until the Big-Bang reforms of 1986, financial markets were governed by a set of self-regulatory organizations,

including stock exchanges, and accountants. Similarly, the free professions were selfregulating without explicit government oversight. The major task of self-regulators, and a task they typically take up on their own initiative, is to regulate their own members. However, a number of self-regulators have also taken up the task of regulating third parties. A good example is stock exchanges vis-à-vis companies that wish to be listed on the exchange. With regards to the effect they have on third parties, SROs resemble in important ways state regulators. The emergence of third party regulation by selfregulators has frequently been a response to the lack of effective enforcement of existing law and legal institutions – i.e. to the same problem that has given rise to the emergence of state regulators.<sup>61</sup>

In the UK, the legislature has increased over time indirect and informal control over SROs. For the most part this has taken the form of threats to establish more stringent regulatory oversight and making this threat credible. In addition, the British legislature has frequently enacted standards that competed with those set by SROs and thereby forced them to raise their own standards and oversight. With respect to enforcing disclosure obligations for listed companies, for example, statutory law has greatly expanded the scope of mandatory disclosure requirements and exempted only companies that complied with higher standards established by the stock exchanges.<sup>62</sup>

By contrast, there are only few formal external control rights over self- or state regulators. English common law courts have exercised considerable self-restraint in

<sup>&</sup>lt;sup>61</sup> This paper does not attempt to analyze the difference between self-regulators and state regulators. However, we would like to point out that self-regulation is an enforcement mechanisms that develops when other enforcement mechanisms are weak or not available. See (Bernstein, 1992; Greif, Milgrom, and Weingast, 1994). For a discussion of the limits of self-enforcement, see (Charny, 1990) and for the importance of the shadow of the state in limiting the danger of conflict of interest, see (DeMarzo, Fishman, and Hagerty, 2000). <sup>62</sup> See (Pistor et al., 2002) for a discussion of the 1927-8 and 1948 company act reforms in the UK.

reviewing regulatory actions – although this may be changing with the expansion of the regulatory state in the UK. Courts have usually taken the view that it is for the parliament to decide to whom to allocate lawmaking and law enforcement powers. The courts would respect this allocation and would only limit the scope of their judicial review to the observance of fundamental principles of due process, in particular the *ultra vires* principle. 63 The limited role of courts in reviewing regulatory actions may come as a surprise to those who believe that common law systems allocate much greater residual lawmaking powers to courts than civil law countries.<sup>64</sup> Yet, English courts have respected the primacy of the parliament's power to realize its own claim to original lawmaking by enacting statutory law or by allocating residual lawmaking and law enforcement powers to different agents. English courts for the most part have therefore limited their role to protecting individuals from infringements of basic legal principles. Courts have drawn the limit where the legislature has tried to curtail even this limited scope of judicial review. Modern acts often state that the decision of the minister in charge shall be final or explicitly limit judicial review. However, in the case Anisminic Ltd. V. Foreign Compensation Commission, the House of Lord has held that such clauses do not necessarily exclude judicial review. 65 Given the long tradition of extensive judicial review by courts in England, the court argued, any restriction had to be interpreted in the

-

<sup>&</sup>lt;sup>63</sup> A distinction is made between substantive and procedural errors, where a substantive violation of the ultra vires principle is found when decisions were made in bad faith or on the basis of false evidence. By contrast, a procedural violation occurs when the procedural requirements of the law are ignored, in particular when the administration ignores the right to be heard or notification of parties whose legal rights may be constrained (Schwarze, 1992)

<sup>&</sup>lt;sup>64</sup> See, for example, (La Porta et al., 2001).

<sup>&</sup>lt;sup>65</sup> 1 All ER 208, [1969] 2 WLR 163, (44 CLJ 126). The issue at hand was the determination of compensation for expropriation of property effectuated by a foreign power with which Britain had resolved the matter by bi-lateral treaty. The relevant provision of the 1950 code (section 4(4)) stated that "The determination by the commission of any application made to them under this Act shall not be called in question in any court of law."

narrowest terms and any exclusive authority to review legislative or administrative actions that was allocated to a special administrative tribunal was to be interpreted to not undermine the review powers of the general courts.

The fact that the legislature has attempted to limit rather than encourage judicial review over regulatory action is interesting to note. It suggests that in its view courts do not play a crucial role as agents of governance over regulators. It may also be interpreted to mean that the legislature has other means to control regulators. <sup>66</sup> The most important devices include the review of administrative decisions by tribunals inside the administration and the strengthening of parliamentary oversight in the form of a new institution established in 1967, the Parliamentary Commissioner for Administration.<sup>67</sup> Yet, even these control mechanisms are weak by design. The commissioner is appointed by the prime minister and may act only when a member of parliament refers a matter to him. In other words, the commissioner facilitates internal review, but does not subject the administration to enhanced external scrutiny or formal legal controls that could be invoked by outsiders. As such, it is fully consistent with the internal control model of British administrative law. Most commentators suggest that in the traditional British system, these means consisted primarily of informal governance, <sup>68</sup> including vertical controls as well as dialogue and cooperation.<sup>69</sup>

<sup>&</sup>lt;sup>66</sup> Pablo Spiller suggests that in light of the restrained role of English courts, contracts rather than licenses have been used in the relation between regulators and the regulated (Levy and Spiller, 1994; Spiller, 1996). We differ in our analysis in that the focus of our research are not the instruments used by state agents vis-àvis private agents (i.e. administrative acts or contracts), but the governance of regulators themselves.

<sup>&</sup>lt;sup>67</sup> Parliamentary Commissioner Act, 1967. <sup>68</sup> (Moran, 2001; Vogel, 1986)

<sup>&</sup>lt;sup>69</sup> [ADD Black.]

Meanwhile, the UK has witnessed the "rise of the regulatory state". 70 This trend entailed the replacement of traditional self-regulatory bodies with new state regulators, many of which are now designed as independent regulators. Thus, financial markets are governed by the Financial Services Authority, food safety is regulated by the new Food Standards Agency (created in response to the BSE crisis), and pharmaceuticals are controlled by the Medicines Control Agency. 71 The key question the UK faces today is. whether the governance mechanisms that have been developed for a different model will be effective in ensuring compliance by increasingly independent regulators with parliamentary objectives. The first response to the new governance challenge has been a strengthening and increasing formalization of internal controls, so much so that books have been written about the rise of "regulation inside government". 72 Our theoretical analysis predicts that the creation of independent regulators should increase the use of external as opposed to internal control mechanisms. Observations in the literature suggesting that the number of court cases is increasing and that courts have enhanced judicial scrutiny of regulatory action<sup>73</sup> are consistent with this prediction. It will be particularly interesting to watch, whether the legislature's perception of the role of judicial review of regulatory actions carried out by the new independent regulators will be changing.

#### France

(Moran, 2001; Moran, 2002).

 <sup>70 (</sup>Moran, 2001).
 71 (Gilardi, 2002; Moran, 2001).

Hood et al (1999) at p. 23 estimate that by 1995 the regulation of government involved at least 135 separate regulators with 14,000 staff and cost over £700 per month to run. CHECK

Just as in the UK, in France regulatory tasks are carried out by agents within the existing state bureaucracy. Unlike in the UK, this is the result not so much of delegation of regulatory powers by the parliament, but of the exercise of genuine lawmaking powers inside the executive. The French constitution of the Fifth Republic gives parliament only limited original lawmaking powers. Only those issues that fall within the domain of "laws" can be determined by act of parliament. All issues not specifically enumerated in the relevant provision of the constitution (Art. 34) fall within the prerogative of the executive, which implements its policies through regulatory decrees and ordinances (Art 37). Moreover, the government has a strong hand in determining parliament's legislative actions not only by initiating bills, but also by virtue of the fact that governmental proposals take precedence over bills proposed by private members of the parliament. In other words, it is not the parliament that delegates regulatory tasks, but it is the executive that has original lawmaking power in areas that the constitution deems not to be "laws".

Consistent with the hierarchical model described above, the control mechanisms that were established are primarily internal. This reflects a strict separation of the executive and the judiciary that was established after the French revolution. The courts were explicitly denied the power to review administrative acts.<sup>75</sup> The constraints on judicial

-

<sup>&</sup>lt;sup>74</sup> Art. 37 states explicitly that all other matters are regulatory in nature("Les matières autres que celles qui sont du domaine de la loi ont un caractère réglementaire. See also (Schwarze, 1992).

<sup>&</sup>lt;sup>75</sup> See Law (relating to judicial organizations) of August 16-24, 1790, Art 13: "Les functions judicaires sont distinctes et demeureront toujours séparées des fonctions administratives. Les juges ne pourront, à peine de forfaiture, troubler, de quelque manière que ce soit, les opèrations des corpos administratifs, ni citer devant eux les administrateurs pour raison de leurs fonctions." [The functions of the judiciary are distinct from and shall be always separate from administrative functions. The judges shall not, at the threat of forfeiture, interfere in any way whatsoever, with the operations of the administrative corps, and may not require administrators to justify their functioning in front of the courts]. As quoted in (Schwarze, 1992) p. 100 note 3. The denial of judicial review of executive acts has a long history in France, which predates the revolution. Already in the early 15<sup>th</sup> century were these disputes resolved in front of quasi-judicial

lawmaking and law enforcement powers thus went much further with respect to administrative than private law. As discussed above, the code civil leaves at least implicitly extensive lawmaking powers in the hands of the courts by using broad and ambiguous language, especially in the area of tort law. By contrast, the review of administrative decision-making, rules and regulations is left to a special body, the Conseil d'Etat, a unique invention of the French system, which was established in 1799. Originally, the *Conseil* did not have any adjudicative functions, but was designed as a consultative body to the government. Only in 1872 was a special litigation section created. Civil servants rather than judges serve even in this section. Nevertheless, today it is widely acknowledged that the litigation section of the Conseil functions largely as an independent administrative court. 76 In addition to reviewing administrative rules and regulations ex post, the Conseil also review proposed bills ex ante. Every bill introduced in parliament must be submitted to the *Conseil d'Etat* for advice. <sup>77</sup> In this case, the advice is not binding, but the procedure nevertheless, gives this institution some ex ante control over lawmaking. Moreover, the *Conseil* must be consulted about delegated legislation designed to fill gaps left in the relevant parent legislation.<sup>78</sup>

Given the extensive *original* rulemaking powers of the executive, it is not surprising to find that administrative decision making is not governed by a comprehensive statutory law. Although a trend to codify areas of administrative law has been noted, <sup>79</sup> these codifications typically organize existing rules, regulations, and case law in a systematic

administrative organs rather than in the general courts. It has been suggested that the doctrine of the separation of power, which gained new currency after the French revolution lent this old tradition new theoretical justification rather than constituting a break with the past. See (Woehrling, 1985).

<sup>&</sup>lt;sup>76</sup> (Schwarze, 1992)
<sup>77</sup> (Obe and Bell, 1993) at p. 61.

<sup>&</sup>lt;sup>78</sup> Ìbid at p. 62.

fashion, but do not add much. As a result, much of the French administrative law is case law as developed by the *Conseil d'Etat*. Its decisions are considered binding on the administration and deviations are "punished" with nullification of the relevant acts.

Aggrieved citizens can bring actions against the state administration. The first step is usually to apply for internal administrative review. Then, the matter will be mediated by a government appointed mediator, an office which was created in the early 1970s. If this does not resolve the matter, it can be referred to *tribunaux adminstratifs*, which since 1953 law have functioned as administrative courts of first instance. The *Conseil d'Etat* can be called upon to review the legality of their decisions in an appellate procedure.

In summary, in France the executive has extensive original lawmaking powers. The major constraints on the use of these powers are civil and political rights enshrined in the constitution. The major function of administrative law as developed in the *Conseil d'Etat's* jurisprudence is the supervision of the functioning of the state apparatus. The supervisory function is not carried out by the legislature, nor by the courts, but by a special agent, the *Conseil d'État*, which itself forms part of the central state apparatus, but functionally separate from it. As other countries in Europe, France is currently contemplating outsourcing major regulatory activities to independent regulatory bodies outside the traditional state administration. In principle, however, this should not affect the role of the *Conseil d'État* as the major oversight body. [CHECK]

Germany

<sup>&</sup>lt;sup>79</sup> Examples include the Code de l'administration communale, the Code de l'environment, and the Code the l'urbanism (ibid at p. 104). See (Schwarze, 1992).

Under the Weimar Constitution of 1918, Germany's first democratic constitution, parliament was vested with the primary lawmaking power. The extent to which the legislature could delegate lawmaking to the state administration was disputed. The German Supreme Court of the time (*Reichsgericht*), however, did not object to farreaching delegation. The only conditions were that the delegation had to be "somehow limited in duration, subject matter, and persons affected" – conditions that were widely regarded as being too lax to be meaningful in practice. <sup>80</sup> The parliament made quite extensive use of the right to delegate lawmaking to administrative agencies in a series of "empowerment statutes" (*Ermächtigungsgesetze*), which were limited in duration, but granted sweeping powers to the executives to address the needs of the people during the Republic's numerous political and economic crises. One of the last these delegating statutes brought and end to the Weimar Republic by empowering the Hitler government to take over essentially all lawmaking functions.

In light of this history, it is not surprising that Germany's new constitution, the 1949 Basic Law of the Federal Republic of Germany explicitly constraints the right of the legislature to delegate lawmaking and the scope of the administration's lawmaking powers. In principle, all acts that may constitute an infringements of the civil and political rights enshrined in the *Grundgesetz* (GG) must be based on a statute passed by parliament. Such a statute may authorize the federal government, a federal minister, or the governments of the Länder to issue regulations. Note, that independent regulatory agencies are not mentioned. Moreover, the statute must determine the content, purpose, and scope of such authorization and the regulation itself must specify its statutory basis. Finally, if the statute provides that the authority may be delegated further (i.e. to lower

<sup>&</sup>lt;sup>80</sup> (Kischel, 1994) at p. 229.

echelons of the administration), this delegation may be done only by regulation, not internal orders.

The system as set up by the constitution thus is one that requires the legislature to be fairly specific when delegating regulatory functions to the administration – a control device we have associated with horizontal governance models. In light of Germany's experience with excessive parliamentary delegation of rule making activities, this is not surprising. At face value, however, it limits the administration's scope of discretion as well as its ability to respond to changes in the environment without explicit legislative authority. Not surprisingly, the system has come under attack in areas such as environmental law, which are more fluid and require greater responsiveness by agents who hold residual lawmaking powers.<sup>81</sup>

In reality, however, there is more flexibility in the system than the constitutional set up might suggest. Whether a statute does clearly circumscribe content, purpose and scope of future regulations has become subject of much litigation and judicial review. This supports our proposition that inherent incompleteness of law limits the ability of legislatures to use the delegating law as a major constraint on regulatory activities. In resolving disputes over the scope of regulatory authority when the law is incomplete, the supreme administrative court relies on a theory of "essentialness" (*Wesentlichkeits-theorie*), which asks whether the specification in the law are sufficient to make the contents of the regulation predictable. This, of course, is a rather open ended standard. Moreover, courts use different standards depending on the nature of the issue that is

\_

<sup>82</sup> (Kischel, 1994) (at p. 233).

<sup>&</sup>lt;sup>81</sup> See (Schwarze, 1992). For a different critique, focusing on the lack of judicial review and citizens' participatory rights, see (Rose-Ackerman, 1995).

being regulated. Where basic civil or political rights are involved, the standards are more stringent than where this is not the case.

Administrative acts are subject to judicial review by the administrative courts after the internal administrative review mechanisms have been exhausted.<sup>83</sup> The focus of administrative judicial review is on the protection of citizen's individual rights. Standing is limited to claims that a particular act violates the plaintiff's individual rights. In the course of the review administrative courts may, however, strike down regulations enacted by the administration that were not sufficiently authorized by statutes passed by the legislature. In fact, this is a quite frequent occurrence, especially when compared to similar cases in France or the US.<sup>84</sup>

Germany has also followed the recent European trend of establishing specialized agencies and allocating to them specific regulatory tasks. An example is the federal commission for securities, which was established in 1994. The powers of these agents, however, are relatively limited. Moreover, they often work directly under the supervision of ministries, just as other parts of the state bureaucracy. Existing constitutional constraints on the delegation of rule making authority effectively limit the ability of the legislature to vest independent regulators with greater rule making authority. Similarly, enforcement activities that go beyond the revocation of licenses and the like typically require endorsement by the judicial apparatus.

## Canada

-

84 (Kischel, 1994) ADD.

<sup>&</sup>lt;sup>83</sup> Internal review must precede litigation. It is typically carried out by the direct superior of the official who has issued the original act. For details, see [ADD].

Canada follows closely the European model in the governance structure of regulatory functions. Only few constraints limited the scope of delegated lawmaking historically. A law on this matter was passed only after World War II. Before then, delegation faced few formal legal constraints<sup>85</sup> An important feature of the Canadian system is that draft regulations are reviewed by the Privy Council Office prior to their adoption to ensure consistency with existing statutory law and the constitution. 86 In 1986 a special secretariat for regulatory affairs were established in the Privy Council Office and a minister responsible for internal affairs was appointed, enabling the Privy Council to exercise greater scrutiny over regulatory rule making. In addition, regulatory acts are subject to judicial review and Canada has a well developed body case law on the appropriate scope of delegating lawmaking powers to administrative agents. Delegated lawmaking is not only a matter of federal law, but takes place also at the state and municipal level. In Quebec, for example, the first law on delegated lawmaking was enacted in 1986 – much later than in other provinces. 87 It uses a very broad and abstract definition of regulation. The most important control mechanism seems to be ex ante control of proposed regulation by the Ministry of Justice. Only after it has been approved and published does a regulation come into effect. In this age of increased regulatory lawmaking it would be interesting to review regulatory practices in different Canadian provinces and to analyze how governance structures used in Quebec compare with

<sup>&</sup>lt;sup>85</sup> Historically, an interesting feature of the Canadian system has been that parliament delegated legislative powers first to the judiciary (the magistrates) rather than the executive. Examples include the 1792 Act to Prevent Accidents by Fire, the 1797 Act for the Regulation of Ferries. See (Holland and McGowan, 1989). The authors suggest that "this would be considered both unusual and improper today" (at p. 8).

<sup>86</sup> CHECK for origins of this feature.

<sup>&</sup>lt;sup>87</sup> The section relies primarily on (Holland and McGowan, 1989).

governance structures employed in other provinces and how they may affect the quality and outcome of such regulation.

As a general matter, however, the increasing importance of regulation in most modern economies suggests that the traditional separation of common law and French civil law may become less decisive over time. As should be clear from the analysis of the three European jurisdictions above, the regulatory governance devices might differ in details, but their basic typologies are quite similar across jurisdictions, and more importantly, across legal families. The major unifying factors were that regulatory functions were assumed by already existing administrations. The creation of "independent" regulators is only a fairly recent phenomenon. The main governance structure that still dominates are hierarchical controls supplemented by judicial review. Yet, the form of hierarchical controls and the scope of judicial review differs from country to country. To curtail the regulators' discretionary powers such a system may create, the UK has used primarily informal control mechanisms – although this system is increasingly giving way to greater formalization and an expansion of intra-government regulation. France created a special internal governance organization staffed with civil servants. Over time, this internal control apparatus has assumed extensive judicial-like review functions to ensure the proper functioning of the state administration. Germany uses a combination of legislative oversight, which is mandated by the constitution, informal controls, and judicial review, which albeit somewhat narrow in scope has been used quite effectively in the past to overturn regulatory actions, including regulatory lawmaking. Still, when compared with the US model, the similarities across these three countries outweigh their differences. As discussed, there are signs that this is changing

and that the UK in particular is in a process of not only outsourcing more regulatory tasks to independent regulators, but also of enhancing external controls, in particular judicial review as internal control devices have become exceedingly costly.

### **United States**

In the United States, the rise of the regulatory state is most closely associated with the New Deal under the Roosevelt administration, even though a number of independent agencies had been established earlier, including the Interstate Commerce Commission of 1887, or the Federal Trade Commission in 1914. 88 What made the New Deal different was a new political philosophy about the function of law and the proper allocation of lawmaking functions within existing constitutional constraints. Whereas traditionally the primary function of law had been to enforce existing property rights, law was now to be used also as an instrument of re-distribution.<sup>89</sup> In areas where prevention rather than redistribution was the primary policy objective, technocrats with specialized expertise rather than judges were given the power to specify, interpret, and enforce these policy objectives. 90

The regulatory system that was designed at the time resembles in many ways the ideal type of a horizontal governance structure of regulators described above. Independent regulators cannot be controlled with the same devices that work quite effectively in a hierarchical setting, such as reward and punishment – be they formal or informal - within clearly established vertical lines of control. Initially, it was hoped that intrinsic control devices, in particular the professionalism and technocratic expertise of

89 (Sunstein, 1987).

<sup>&</sup>lt;sup>88</sup> For an overview of the history of regulation in the US, see (Stewart, 1975).

regulators would suffice to ensure compliance with the legislated mandate.<sup>91</sup> Thus, legislative controls were often limited in the delegating legislation. This, however, proved difficult. For some regulators their powers were clearly circumscribed in the law. Many regulators, however, were given broad lawmaking and law enforcement powers.

Creating independent agencies outside the existing state bureaucracy immunized these agents to some extent from control by the executive. At least in part this was a response to existing constitutional constraints as interpreted by the Supreme Court. The US constitution vests Congress with lawmaking powers. In fact, when striking down the National Industry Recovery Act in 1933 and 1935 the Supreme Court held that Congress could not freely delegate lawmaking powers. At the very least, it had to set the relevant standards itself. These decisions notwithstanding, the powers of regulators were greatly expanded over time. This is true in particular in areas we are most interested in, i.e. regulations aimed at protecting the public from externalities arising from potentially dangerous operations or the use and emission of dangerous substances Some laws mandated pre-approval obligations, others required regular review of regulatory performance in lawmaking and law enforcement. However, these legislative control mechanisms have not been used very extensively.

---

<sup>&</sup>lt;sup>90</sup> See also (Stewart, 1975).

<sup>&</sup>lt;sup>91</sup> This belief found one of its strongest advocates in (Landis, 1938).

<sup>&</sup>lt;sup>92</sup> Panama Refining Co. v. Ryan, 293 U.S. 388 (1933); A.L.A. Schlechter Poulty Corp. v. United States, 295 U.S. 495 (1935).

<sup>&</sup>lt;sup>93</sup> Sunstein suggests that the courts simply succumbed to political pressure that demanded regulatory intervention [ADD]. An alternative interpretation would be that the delegation of lawmaking powers to independent agents was perceived to be less problematic as it did not challenge the separation of powers under the constitution.

<sup>&</sup>lt;sup>94</sup> (Epstein and O'Halloran, 1999). See in particular Table 5.4 "Acts with Lowest and Highest Delegation Ratio. In the first category is tax law, social security law, and civil rights act; in the latter the water quality act, the clean air act, and the air quality act, among others.

<sup>&</sup>lt;sup>95</sup> (McCubbins, Noll, and Weingast, 1987). See also Epstein and O'Halloran (1999) pp. 21. Note, however, that other studies have shown that political control by Congressional committees over regulatory action can

constraints on regulatory agencies increasingly substituted for substantive delimitation of administrative action.

This procedural governance structure was institutionalized with the adoption of the Administrative Procedure Act in 1946. The Act requires regulators to ensure public participation, give reasons for the acts they take, and ensure proper review both inside the administration, and ultimately judicial review. The Act thereby formalizes multiple channels of accountability. Given the regulators' independence, vertical control was not an option and monitoring by Congress alone proved costly to be effective. Horeover, limiting the scope of regulatory powers by writing a highly specific law faced the problem of incompleteness of this law. Even where possible, it impeded the regulator's ability to act promptly and respond flexibly to a changing environment – a major benefit of lawmaking by regulators as opposed to legislatures. The use of multiple avenues of accountability, including particular participation rights and judicial review implied that Congress did not have to carry the burden of regulatory oversight alone. Instead, the responsibility for regulatory governance is shared with those subject to regulation who may challenge in court the failure by regulators to comply with the established constitutional constraints. Horeover, and the responsibility for regulatory governance is shared with those subject to regulation who

The system is by no means perfect. Reigning in overzealous regulators is not always easy, nor have regulators always lived up to the task of effective law enforcement, as only recently witnessed in the world of finance. 98 Arguably it is at least as vulnerable

be quite powerful. Weingast and Moran (1983), for example show that Congress through specialized sub-committees exercises substantial control over the Federal Trade Commission

<sup>&</sup>lt;sup>96</sup> The problem of monitoring costs is stressed by (McCubbins, Noll, and Weingast, 1987).

<sup>97 (</sup>McCubbins, Noll, and Weingast, 1987) argue extensively that the procedural devices were used to limit the cost of active Congressional monitoring and oversight.

<sup>&</sup>lt;sup>98</sup> Major actions were brought by New York's General Attorney, Eliot Spitzer, rather than by the then chairman of the Securities and Exchange Commission (SEC), Harvey Pitt. See Andy Serwer and Julie

to failure – i.e. under- or over-enforcement when compared to socially optimal levels - than the hierarchical models described above. Yet, such hierarchical models were not feasible given existing constitutional constraints in the US, in particular the non-delegation doctrine.

The scope of judicial review has been subject to much controversy, both among courts and academics. Particularly problematic is the extent to which courts may substitute their own judgment in substantive matters for that of regulators. In exercising judicial review, US courts have at least originally exercised less self-restraint than British courts. This appears to be at least in part a function of the different constitutional framework in which they operate. British courts operate in a system that gives parliament control over the executive and thus is less adverse to delegating lawmaking to the executive. By contrast, US courts operate in a system that gives democratic legitimization to both the president and Congress, and vests one with executive powers, the other with lawmaking powers. Not surprisingly, US courts have scrutinized the substantive justification for regulatory intervention to a greater degree under the "hard look doctrine", which imposes on regulators the need to justify their intervention with detailed cost benefit analyses. Courts have also substituted the regulator's judgment with their own. A departure from this – in comparative perspective quite extensive – judicial review has been the Chevron case decided by the Supreme Court in 1984. 99 In this case, the Supreme Court held that courts could not simply replace the regulator's judgment with

C.

Schlosser, Wall Street Story: There's a Rumbple Between Harvey Pitt and Eliot Spitzer. Is Wall Street Big Enough for Both of Them? *Fortune,* June 8 2002 at p. 28.

<sup>&</sup>lt;sup>99</sup> 467 U.S. 837 (1984). For a detailed analysis of this case and its aftermath, see (Merrill and Hickman, 2001).

their own. To the extent Congress has vested regulators with powers and they act in accordance with these powers, courts should defer to administrative competence.

In the twenty years since Chevron the governance structure of regulators in the United States has undergone further change. One can be described as the regulators' flight into informality, the other as the strengthening of presidential control. The horizontal governance structure of the US system described above has imposed substantial costs on the regulators. In particular, it has considerably slowed down the process of rulemaking, thereby undermining one of the greatest advantages of a regulator, namely to flexibly adapt existing rules to a changing environment. In response, regulators have taken flight into informality. Rather than expounding explicit rules, agencies have increasingly used informal non-binding guidelines to suggest what actions they might take in case of violations. At the same time, the flight into informality has raised the specter that regulators may once again escape accountability.

The second trend is the rise of the "presidential administration". The clearest sign has been the creation of the Office of Management and Budget (OMB) during the first Reagan administration. In the terms of analysis, OMB resembles a superregulator. All agencies are required to submit to the OMB's Office of Information any proposed major rule for pre-publication review. This allows OMB to take proactive steps in enjoining regulatory lawmaking before it is being implemented. Each submission has to be accompanied by a regulatory impact analysis, which must give a cost benefit analysis of the chosen strategy, allowing OMB to exercise substantive oversight over regulatory rulemaking. Subsequently, agencies were even required to present rulemaking

1/

<sup>&</sup>lt;sup>100</sup> (Kagan, 2001). See also (Sunstein, 1987) making a strong case for enhanced presidential oversight over regulatory actions.

plans for forthcoming years, allowing the presidential administration to directly influence this process.

This development is transforming the governance over regulators from a horizontal to a more vertical structure. Multiple horizontal control mechanisms may have ensured that regulatory powers were not misused. However, they have come at substantial costs. Moreover, they created a control vacuum with respect to setting regulatory policy agendas, which the presidential administration has gradually filled. The transformation of the US model into a more hierarchical governance structure is by no means complete. The above analysis only indicates a trend. The Presidential administration has not replaced judicial review, public participation, or Congress's power to review the scope of delegated regulatory tasks. The trend does, however, suggest that the horizontal model is costly and difficult to sustain – important lessons especially for countries that are only now witnessing the rise of the regulatory state. Regulatory accountability is a much more complex task in this model than in the vertical control model as it requires the design of proper incentives for different agents that are expected to participate in the governance task. Even where it succeeds in effectuating negative control against abuse, it seems vulnerable to the usurpation of positive control by different actors – be they part of the government (i.e. the president) or private agents (i.e. powerful interest groups).

#### 7. Conclusion

<sup>&</sup>lt;sup>101</sup> (Sunstein, 1987) pp. 454 and (Kagan, 2001) pp. 2277.

In recent years much has been made of the difference between common law and civil law countries in determining financial market development, <sup>102</sup> good governance, <sup>103</sup> and ultimately economic growth. 104 Most of these empirical studies concluded that common law countries were superior to civil law countries and have related these findings to institutional differences, in particular to the role of courts and their relation to the government. 105 Over the past century, however, agents other than courts have come to play a crucial role in lawmaking and law enforcement. We argue that the rise of the regulatory state is a response to the inability of the traditional law enforcement agency, the courts, to effectively deter socially harmful actions. We observe the rise of regulatory functions in different jurisdictions irrespective of their legal origin. However, we do note that different countries employ different governance strategies to mitigate the problems that arise when extensive lawmaking and law enforcement powers are placed in the hands of a single, albeit specialized agent. While there are variations in detail, we single out two ideal types of regulatory governance, a hierarchical and a horizontal one, noting that the jurisdictions we have analyzed increasingly combine features of both to address the tasks posed by the rise of the regulatory state.

Recall that the starting point of our analysis was that just as contracting parties are unable to write complete contracts, so lawmakers are unable to write complete laws. Unless lawmakers allocate some residual lawmaking powers to others, in particular to agents in charge of enforcing the law, law enforcement is likely to be ineffective as

<sup>&</sup>lt;sup>102</sup> (La Porta et al., 1997; La Porta et al., 1998) ... (La Porta et al., 1999)

<sup>104 (</sup>Mahoney, 2001). 105 (Glaeser and Shleifer, 2001).

technological and socioeconomic change will render existing law exceedingly incomplete and offer new arbitrage opportunities for agents to escape the reach of the law.

Traditionally, the most important agents that exercised residual lawmaking powers were courts. This is true for all countries analyzed, irrespective of legal origin. Over time, however, regulators have come to play an increasingly important role as residual lawmakers and law enforcers. We argue that their major advantage over courts is that unlike courts they can enforce law proactively and that they can make rules ex ante, which enhances their ability to prevent harm as opposed only remedying harm after it has been done.

Using regulators to optimize law enforcement to socially desirable levels assumes that regulators are social welfare maximizers, or at the very least, that they share the same objectives as legislatures. In the real world this is unlikely to be the case. The key question then become how the principals who delegate lawmaking and law enforcement powers to regulators can make sure that regulators stay as close to their objectives as possible. In theory they have two options: they may use hierarchical internal controls or they may use horizontal devices (more akin to contracts than commands) and decentralized mechanisms of accountability to control the regulators. Effective hierarchical or internal controls are available only in systems where the legislature has real leverage over the state administration. Even then, the efficacy of this model may be challenged by the growth of regulatory tasks, both in number, and in complexity. Alternatively, the legislature may attempt to write a highly complete law to control regulatory discretion and monitor compliance either itself or, by delegating monitoring to others, including to those that are being regulated. Limiting regulatory discretion by

writing a complete law is not feasible for reasons explained. Moreover, keeping regulators on a short leash constraints their ability to make the most of their proactive lawmaking and law enforcement powers. Other governance devices must be sought to retain the benefits of a flexible regulator while ensuring optimal levels of compliance. One of the most important monitoring devices used in the horizontal system has been judicial review initiated by affected individuals, as well as extensive citizen participation in lawmaking processes. By contrast, hierarchical governance structures rely primarily on internal controls and make comparatively less use of judicial review.

Our comparative analysis suggests that both ideal regulatory regimes – the hierarchical or internal model on the one hand, and the horizontal, or external governance model on the other – create their own costs. Reforming and adapting the governance structure is an ongoing process in all countries, albeit arguably to a greater extent in the US and the UK than – so far - in either Germany or France. In adapting their existing governance structure, each country faces the challenge of finding new solutions given its own constitutional constraints. As Levy and Spiller<sup>106</sup> have pointed out, what appears to be important is the "fit" of the chosen governance structure, not its inherent superiority.

# References:

Banner, Stuart. 1998. Anglo-American Securities Regulation- Cultural and Political Roots, 1690-1860. Cambridge: Cambridge University Press.

Page 1. Ulright 1992. Pick Society Towards a New Modernity. Landon: Saga

Beck, Ulrich. 1992. Risk Society: Towards a New Modernity. London: Sage.

Becker, Gary S. 1968. Crime and Punishment: An Economic Approach. *Journal of Political Economy* 76:169-217.

106 (Levy and Spiller, 1994)

- Bernstein, Lisa. 1992. Opting out of the Legal System: Extralegal Contractual Relations in the Diamond Industry. *Journal of Legal Studies* 21 (1):115-157.
- Boudreaux, Donald J., and A.C. Pritchard. 1994. Reassessing the Role of the Independent Judiciary in Enforcing Interest-Group Bargains. *Constitutional Political Economy* 5 (1):1-21.
- Bratton, William. 2002. Enron and the Dark Side of Shareholder Value. *TUlane Law Review* 76.
- Brickman, Ronald, Ehila Jasanoff, and Thomas Ilgen. 1985. Controlling Chemicals: The Politics of Regulation in Europe and the United States. Ithaca and London: Cornell University Press.
- Charny, David. 1990. Nonlegal Sanctions in Commercial Relationships. *Harvard Law Review* 104:373.
- Coase, Ronald H. 1960. The Problem of Social Cost. *Journal of Law and Economics* 3:1-44.
- Coffee, John C. Jr. 1984. Market Failure and the Economic Case for a Mandatory Disclosure System. *Virginia Law Review* 70:717-753.
- David, Rene, and John E. Brierly. 1985. *Major Legal Systems in the Word Today*. London: Steven & Sons.
- <u>DeMarzo, Peter M., Michael J. Fishman, and Kathleen M. Hagerty. 2000. Contracting and Enforcement with a Self-Regulatory Organization: Research Seminar in Law and Economics.</u>
- Epstein, David, and Sharyn O'Halloran. 1999. *Delegating Powers A Transaction Cost Politics Approach to Policy Making under Separate Powers*. Cambridge: Cambridge University Press.
- Fiorina, Morris P. 1986. Legislator Uncertainty, Legislative Control, and the Delegation of Legislative Power. *Journal of Law, Economics and Organization* 2 (1):33-51.
- Giddens, Anthony. 1990. The Consequences of Modernity. Cambridge: Polity.
- Gilardi, Fabrizio. 2002. Policy Credibility and Delegation to Independent Regulatory

  Agencies: A Comparative Empirical Analysis. *Journal of European Public Policy* 9

  (6):873-893.
- Glaeser, Edward, and Andrei Shleifer. 2003. The Rise of the Regulatory State. *Journal of Economic Literature* (June): forthcoming.
- Glaeser, Edward L., and Andrei Shleifer. 2001. Legal Origins. *Quarterly Journal of Economics*.
- Glendon, Mary Ann, Michael W. Gordon, and Christopher Osakwe. 1994. Comparative

  Legal Traditions: Text, Material and Cases on the Civil and Common Law

  Traditions, with Special References to French, German, and English. St. Paul, Minn.:

  West.
- Goldberg, Victor. 1976. Regulation and Administered Contracts. *Bell Journal of Economics* 7 (2):426-452.
- Greif, Avner, Paul Milgrom, and Barry R. Weingast. 1994. Coordination, Commitment, and Enforcement: The Case of the Merchant Guild. *Journal of Political Economy* 102 (4).
- Grossman, Sanford J., and Oliver D. Hart. 1986. The Costs and Benefits of Ownership: A Theory of Vertical and Lateral Integration. *Journal of Political Economy* 94 (4):691-719.

- Hart, H.L.A. 1961. The Concept of Law. Oxford: Oxford University Press.
- Hart, Oliver. 1995. Firms, Contracts, and Financial Structure. Oxford: Clarendon Press.
- Holland, Denys C., and McGowan. 1989. *Delegated Legislation in Canada*. Toronto, Calgary, Vancouver: Carswell.
- Hood, Christopher, Colin Scott, Oliver James, Georges Jones, and Tony Travers. 1999. *Regulation Inside Government: Waste-Wathcers, Quality Police, and Sleaze Busters*.

  Oxford: Oxford University Press.
- <u>Kagan, Elena. 2001. Presidential Administration. *Harvard Law Review* 114 (June):2245-2385.</u>
- <u>Kischel, Uwe. 1994. Delegation of Legislative Power to Agencies: A Comparative Analysis of United States and German Law. *Administrative Law Review* 46:213-256.</u>
- <u>La Porta, Raphael, Florencio Lopez-de-Silanes, Cristian Pop-Eleches, and Andrei Shleifer.</u> 2001. The Guarantees of Freedom: Harvard University.
- <u>La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert Vishny. 1999.</u>

  <u>The Quality of Government. The Journal of Law, Economics & Organization 15</u>
  (1):222-283.
- <u>La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert W. Vishny.</u>

  <u>1997. Legal Determinants of External Finance. *Journal of Finance* LII (3):11311150.</u>
- <u>La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert W. Vishny.</u> 1998. Law and Finance. *Journal of Political Economy* 106 (6):1113-1155.
- <u>Landes, William M., and Richard A. Posner. 1975. The Independent Judiciary in an Interest-Group Perspective. *Journal of Law and Economics* 18:875-901.</u>
- Landis, James McCauley. 1938. The Administrative Process. Harvard University Press.
- Levy, Brian, and Pablo Spiller. 1994. The Institutional Foundations of Regulatory Commitment: A Comparative Analysis of Telecommunications Regulation.
- Mahoney, Paul. 2001. The Common Law and Economic Growth: Hayek Might be Right. Journal of Legal Studies.
- <u>Majone, Ginadomenico. 1994. The Rise of the Regulatory State in Europe. West European Politics 17 (3):77-101.</u>
- Mattei, Ugo. 2000. *Comparative Law and Economics*. Ann Arbor: University of Michigan Press.
- McCubbins, Mathew D., Roger G. Noll, and Barry R. Weingast. 1987. Administrative Procedures as Instruments of Political Control. *Journal of Law, Economics and Organization* 3 (2):243-277.
- Merkt, Hanno. 1997. Zur Entwicklung des Deutschen Börsenrechts von den Anfängen bis zum Zweiten Finanzmarktförderungsgesetz. In *Börsenreform Eine ökonomische, rechtsvergleichende und rechtspolitische Untersuchung*, edited by K. J. Hopt, B. Rudolph and H. Baum. Stuttgart: Schäffer-Poeschel Verlag.
- Merrill, Thomas, and Kristin Hickman. 2001. Chevron's Domain. *The Georgetown Law Journal* 89.
- Michie, Ranald C. 1999. *The London Stock Exchange: A History*. Oxford: Oxford University Press.
- Moran, Michael. 2001. The Rise of the Regulatory State in Britain. *Paliamentary Affairs* 54:19-34.

- Moran, Michael. 2002. Review Article: Understanding the Regulatory State. *British Journal of Political Science* 32 (391-413).
- Obe, L. Neville Brown, and John S.` Bell. 1993. *French Administrative Law*. Oxford: Clarendon Press.
- Oserloh, Margit, and Bruno Frey. 2000. Motivation, Knowledge Transfer, and Organizational Forms. *Institute for Empirical Research in Economics Working Paper* 27.
- Pistor, Katharina, Yoram Keinan, Jan Kleinheisterkamp, and Mark West. 2002. The Evolution of Corporate Law. *University of Pennsylvania Journal of International Economic Law* 23 (4):791-871.
- Pistor, Katharina, and Chenggang Xu. 2003. Incomplete Law. *Journal of International Law and Politics* (forthcoming).
- <u>Polinsky</u>, Mitchell, and Steven Shavell. 2000. The Economic Theory of Public Enforcement of Law. *Journal of Economic Literature* 38 (1):45-76.
- <u>Posner, Richard. 1974. Theories of Economic Regulation. Bell Journal of Economic Management 5:335-358.</u>
- Posner, Richard. 1993. What do Judges Maximize? (The Same Thing Everybody Else Does). Supreme Court Economic Review 3:1.
- Priest, George L. 1993. The Origins of Utility Regulation and the Theories of Regulation Debate. *Journal of Law and Economics* 36 (1):289-323.
- Ramseyer, J. Mark. 1994. The Puzzling (In)Dependence of Courts: A Comparative Approach. *Journal of Legal Studies* 23:721-747.
- Robb, George. 1992. White-Collar Crime in Modern England: Financial Fraud and Business Morality, 1845-1929. Cambridge: Cambridge University Press.
- Romer, Christian D. 1990. The Great Crash and the Onset of the Great Depression. *Quarterly Journal of Economics* 105 (3):597-624.
- Rose-Ackerman, Susan. 1995. Controlling Environmental Policy: The Limits of Public Law in Germany and the United States. New Haven: Yale University Press.
- Salzberger, Eli. 1993. A Positive Analysis of the Doctrine of Separation of Powers, or:

  Why do we have an Independent Judiciary. *International Review of Law and Economics* 13:349-379.
- Schwarze, Jürgen. 1992. *European Administrative Law*. London: Sweet and Maxwell. Seligman, Joel. 1983. The Historical Need for a Mandatory Corporate Disclosure System. *Journal of Corporation Law* 9 (1):1-61.
- Shavell, Steven. 1984. Liability for Harm versus Regulation of Safety. *Journal of Legal Studies* 13 (June):357-374.
- Spiller, Pablo T. 1996. A Positive Political Theory of Regulatory Instruments: Contracts, Administrative Law, or Regulatory Specifity? Southern California Law Review 69:477-515.
- Stephenson, Matthew Calb. 2003. When the Devil Turns .... The Political Foundations of Independent Judicial Review. *Journal of Legal Studies* forthcoming.
- Stewart, Richard B. 1975. The Reformation of American Administrative law. *Harvard Law Review* 88 (1669).
- Stigler, George J. 1970. The Optimal Enforcement of Laws. *Journal of Political Economy* 78 (3):56-536.
- Stigler, Joseph. 1971. The Theory of Economic Regulation. Bell Journal of Economics.

- Strauss, Peter. 1984. The Place of Agencies in Government: Separation of Powers and the Fourth Branch. *Columbia Law Review* 84:573-669.
- Sunstein, Cass R. 1987. Constitutionalism after the New Deal. Harvard Law Review 421.
- <u>Temin, Peter. 1992. The Origin of Compulsory Drug Prescriptions. Journal of Law, Economics and Organization 2 (91-105).</u>
- Thatcher, mark. 2002. Analysing Regulatory Reform in Europe. *Journal of European Public Policy* 9 (6):859-872.
- <u>Vogel, David. 1986. National Styles of Regulation: Environmental Policy in Great</u> *Britain and the United States.* Ithaca: Cornell University Press.
- Weingast, Barry R., and Mark J. Moran. 1983. Bureaucratic Discretion or Congressional Control? Regulatory Policymaking by the Federal Trade Commission. *Journal of Political Economy* 91 (5):765-800.
- Williamson, Oliver. 1979. Transaction-Cost Economics: The Governance of Contractual Relations. *Journal of Law and Economics* 22 (October):233-61.
- Woehrling, Jean-Marie. 1985. Die französiche Verwaltungsgerichtsbarkeit im Vergleich mit der deutschen. *Neuv Zeitschrift für Verwaltungsrecht* 1985 (1):21-27.
- Xu, Chenggang, and Katharina Pistor. 2003. Law Enforcement under Incomplete Law.

  Columbia Law and Economic Working Paper Series No. 222 (available on ssrn.com).
- Zweigert, Konrad, and Hein Kötz. 1984. *Einführung in die Rechtsvergleichung*. 2 vols. Vol. 2. Tübingen: J.C.B. Mohr (Siebeck).
- Zweigert, Konrad, and Hein Kötz. 1998. *Introduction to Comparative Law*. Oxford: Clarendon Press.